Author’s response to reviews

Title: Acute intestinal obstruction due to extrinsic compression by previa myoma and ectopic pregnancy: a case report

Authors:

HARISSOU ADAMOU (harissou1976@yahoo.fr)
Ibrahim Amadou Magagi (imagagi@gmail.com)
Souleymane Oumarou Garba (oumarougarba54@yahoo.fr)
Oumarou Habou (bhomar70@yahoo.fr)

Version: 1 Date: 06 Nov 2017

Author’s response to reviews:

Dear Reviewers of journal of medical case reports,

We thank you for your corrections, we took into account to improve the manuscript.
We have corrected the spelling and syntax errors.

Reviewer 1:
We modified the summary with remarks

Reviewer 2
Through this observation, we want to report the difficulty of the etiological research of an occlusion in women of childbearing age. The patient from a remote environment did not have access to a health facility and did not have previous gynecological consultations. She was admitted to our hospital with an acute bowel obstruction chart. The ruptured ectopic pregnancy and the myomas were of intra-operative discovery. In fact, the clinical picture can be related to both ruptured ectopic pregnancy and massive compressive myomas. However, myoma compression was evident intra-operatively.